# **Research Paper:** Effect of Mirror Therapy on the Motor **3** Recovery in Patients After Stroke: A Randomized Clinical Trial

Sadra Ashrafi<sup>1</sup> 💿, Maryam Shabaani Mehr<sup>2</sup> 💿, Tahereh Khaleghdoost Mohammadi<sup>2\*</sup> 💿, Shirin Jafroudi<sup>2</sup> 💿, Ehsan Kazemnezhad Leyli<sup>3</sup> 💿

1. Student Research Committee, Chronic Kidney Disease Research Center, Shahid Beheshti University of Medical Sciences, Tehran, Iran.

2. Department of Nursing, School of Nursing and Midwifery, Guilan University of Medical Sciences, Rasht, Iran.

3. Social Determinants of Health Research Center, Department of Biostatistics, School of Nursing and Midwifery, Guilan University of Medical Sciences, Rasht, Iran.



**Citation** Ashrafi S, Shabaani Mehr M, Khaleghdoost Mohammadi T, Jafroudi S, Kazemnezhad Leyli E. Effect of Mirror Therapy on the Motor Recovery in Patients After Stroke: A Randomized Clinical Trial Iranian Rehabilitation Journal. 2022; 20(Special Issue):65-78. http://dx.doi.org/10.32598/irj.20.SpecialIssue.1519.1

doi http://dx.doi.org/10.32598/irj.20.SpecialIssue.1519.1

# 

Article info: Received: 01 Jul 2021 Accepted: 17 Nov 2021 Available Online: 01 Jan 2022

#### **Keywords:**

Mirror therapy, Motor recovery, Stroke, Rehabilitation, Disability

# ABSTRACT

**Objectives:** One of the most important problems seen in patients after stroke is that they cannot develop normal muscle strength. In recent years, the use of Mirror Therapy (MT) in the recovery of this condition has been noticed in different studies. This study investigated the effect of MT on motor recovery in patients after stroke.

**Methods:** In this clinical trial, 93 patients were divided into three groups, including MT, nonreflective surface, and control groups. The tools used in this study included the patient's profile questionnaire, Mini-Mental State Examination Test, and Brunnstrom Recovery Stages. After the routine physiotherapy program, the intervention groups underwent MT for 20 sessions. The analysis of data was performed by SPSS software v. 22.

**Results:** There was a significant difference between the non-reflective surface and MT groups (P=0.043) in pairwise comparison of their motor recovery stages in the  $20^{th}$  session, but the difference between the non-reflective surface and control groups was not significant. There was also a significant statistical difference between the MT and control groups in motor recovery stages in the  $20^{th}$  session, (P=0.0332)

**Conclusion:** The obtained findings suggest that MT can increase patients' motor recovery after stroke. This method can be used as a simple, cheap, and usable method at home.

\* Corresponding Author: Tahereh Khaleghdoost Mohammadi, Instructor.

Address: Department of Nursing, School of Nursing and Midwifery, Guilan University of Medical Sciences, Rasht, Iran. Tel: +98 (13) 33555056

.....

E-mail: khaleghdoost@gums.ac.ir

# Highlights

• Stroke patients need simple and accessible rehabilitation methods to improve motor impairment. Mirror therapy is one of these methods.

- In order to control the placebo effect of mirror therapy, a third group, a non-reflective group, was used in the present study.
- The results of this study showed that mirror therapy significantly affected the motor recovery of stroke patients.

• Mirror therapy can be used besides routine rehabilitation programs in the recovery of a patient's motor function. To do this therapy at home, the patients and their families need to be trained by a rehabilitation expert.

# Plain Language Summary

Mirror therapy is a feasible and inexpensive rehabilitation method that can improve the outcomes of routine physiotherapy programs among stroke patients. Moreover, this method is so simple that can be taught to the patients or their families. Thus, they can perform it themselves at home.

## 1. Introduction



troke is the second leading cause of death in the world [1, 2]. Ten percent of 55 million deaths occurring worldwide each year is due to stroke. The number of people who died from stroke increased from 5.29 million (5.22-5.40) to 6.17 million

(6.04-6.33) between 2007 and 2017 [3]. In the United States, From 780,000 strokes occurring each year (one in every 40 seconds), 150,000 people died (one in every 4-3 minutes) [4]. According to the World Health Organization (WHO), there will be a 30% increase in stroke incidence in the EU between 2000 and 2025 [5]. Based on the data collected from 1990 to 2008 in Iran, the annual stroke incidence in various ages ranged from 23 to 103 per 100,000 [6]. A study done in 2016 indicated that the crude annual incidence rate of First-Ever Stroke (FES) was 144 for men and 133 for women in Iran. Totally, stroke occurs in 139 per 100000 people [7].

One of the most important problems seen in patients after stroke is that they cannot develop normal muscle strength, and they also have problems in maintaining normal balance, initiation, and control of their movements. This muscle weakness is due to the reduction in the number of functioning motor units, change of fiber type from low force-producing fibers to high forceproducing fibers, and decrease in motor unit firing rates [8]. The two most common motor deficits in these patients are hemiparesis and hemiplegia [9]. Up to 85% of stroke survivors experience hemiparesis, and 55%–75% continue to have limitations in upper extremity function [10]. About one-third of these patients are dependent on others in most of the Activities of Daily Living (ADL). Therefore, the stroke seems to be one of the most disabling diseases globally [11, 12].

Neurological recovery does not occur at a constant rate at all times after stroke. The maximum rate of recovery is observed in the first three months. However, according to some studies, there are also some degrees of recovery in the months or years later. In the past, researchers believed that much of this recovery was spontaneous, and rehabilitation strategies could not improve patients' condition, but new findings have shown that these strategies could help the brain recover more quickly [13].

The brain's self-recovery results from the brain's effort to rebalance its information systems and reaction mechanisms responding to damage, which is called neuroplasticity. Rehabilitation can help the brain in this procedure [14].

Recently, several studies have introduced new methods in rehabilitating stroke patients, including robotic-assisted training and constraint-induced movement therapy. Unfortunately, there are some problems with these strategies. For example, some of them are expensive. They also often require a person, preferably a specialist, to train the patient continuously. On the other hand, there is another rehabilitation program called Mirror Therapy (MT), which is cheap and can be simply done by the patients themselves [15]. This method was firstly developed by Ramachandran and Roger-Ramachandran to treat patients with phantom limb syndrome after arm amputation in 1996 [16]. In the mentioned study, the patients considered the reflection of the intact arm in the mirror as their amputated arm. MT is based on two concepts: Mirror neurons and neuroplasticity. Mirror neurons are the neurons that not only activate when you are performing an action but also when you are looking at the same action but not performing it [17, 18]. Furthermore, neuroplasticity is the brain's ability to learn new things by changing the ways it uses to deliver signals between neurons [19].

In recent years, the use of MT in the treatment of some complications of diseases, such as complex regional pain syndrome [11, 20, 21] and phantom pain syndrome [22, 23] has been studied. Mazlom et al. [24] investigated the effect of MT on 38 stroke patients in a randomized clinical trial. They found that motor recovery was significantly better in the intervention group. Besides, some other studies have examined the application of this treatment for complications following a stroke. Many of them had some limitations. For example, they reviewed the use of MT in the upper or lower limbs separately [15, 25-27].

Also, the number of samples studied was limited, and it is not clear how many sessions MT should be performed and how many minutes each session should be. Moreover, in none of the previous studies, the possible placebo effect of MT has been assessed. It is, therefore, necessary to investigate the effect of MT on stroke complications in a larger population and evaluate their motor recovery.

It is hoped that the results of this study will provide a way for patients to live better. This study was done to evaluate the effect of MT on motor recovery in patients after stroke and related factors.

# 2. Material and Method

# Study design & participants

In this clinical trial, the effect of MT was evaluated on the motor recovery of patients after stroke. The study was done in the physiotherapy center of Rasht (a city in the north of Iran) Disabled and Elderly Center. Cases with stroke who were referred to this center for rehabilitation in 2016 formed the study population. The sample size required to compare the effects of MT on the motor recovery of patients with stroke was calculated based on the results of the study by Mazlom et al. [24]. The sample size for comparing two groups was determined 66 cases with 99% confidence interval and 95% power in a two-tailed test. According to the below formula, the sample size for comparing three groups was calculated at 93.3, which was considered to be approximately 93 (31 for each group).

For three groups= $n=n_0 \times \sqrt{g}-1=66 \times \sqrt{3}-1=93.3 \approx 93$ 

These people were divided into three groups, including MT, non-reflective surface intervention, and control (routine physiotherapy) groups.

Sampling was performed using block randomization. Accordingly, people referring to the physiotherapy center were first stratified into blocks (block size of 6) according to their Brunnstrom motor recovery stages before the intervention. Then, the people in each block were randomly divided into three groups with a 1:1:1 ratio. The inclusion and exclusion criteria of the study were as follows:

#### **Inclusion criteria**

• Patients who had a history of stroke for the first time, confirmed by a neurologist with Computerized Tomography (CT) scan or Magnetic Resonance Imaging (MRI).

• At least one month should have passed since the patients' stroke because it is only in the subacute and chronic stages that the patients go to rehabilitation centers to continue their treatment.

• The patients should be under the routine rehabilitation program when entering the study

• Their Brunnstrom stages (the index of functional recovery) must be 0 to 3.

• Moreover, their scores based on the mini-mental state examination test should be more than 24. They should not have any underlying motor defects that may interfere with the study, such as myasthenia gravis, multiple sclerosis, etc.

• The patient's specialist physician should have confirmed that the patient does not have any verbal disturbances, previous dementia, globular aphasia, and any visual impairment.

#### **Exclusion criteria**

• If patients for any reason could not participate in the program for more than four intermittent sessions or two consecutive sessions, they were excluded from the study.

• Death of the patient

• The dissatisfaction of the patient or his companions to continue cooperation in the research

The present study was approved by the Ethics Committee of Guilan University of Medical Sciences with the code IR.GUMS.REC.1394.11 and registered at the Iranian Registry of Clinical Trials with the code IRCT201504224787N5. The sampling period was five months (July to November 2016). If the patients referring to the physiotherapy had all the inclusion criteria, they would enroll in the study after giving written consent. For illiterate cases, their companions gave written consent.

Out of 263 patients with stroke who were referred to the center, a total of 93 patients who had inclusion criteria were enrolled in the study. These patients were divided into three groups, two intervention groups (MT and non-reflective surface) and one control group. Of the 93 patients admitted, two were discharged from the MT group (due to absence in physiotherapy sessions and lack of regular visits) and one from the control group (due to death). These patients were replaced by new ones (Figure 1).

#### Measures

The tools used in this study included the patient's profile questionnaire (consisting of two parts: demographic and disease-related information), Mini-Mental State Examination Test (MMSE), and Brunnstrom Staging (the index of functional recovery). The demographic part of the patient's profile questionnaire includes sex, age, height, weight, body mass index, level of education, income, occupation, and marital status. The disease-related part consists of the post-stroke period, the type of stroke, the affected side of the patient's body, the dominant hemisphere, and the muscular tone of the upper and lower limbs. This information was obtained through interviews with the patient or a companion or extracted from the patient profile.

Mini-Mental State Examination Test (MMSE) is a patient cognitive assessment tool that briefly measures patients' awareness of location and time, immediate memory, shortterm verbal memory, calculation, language, and drawing ability. Based on the MMSE, patients are divided into three levels: Absence of cognitive defect (24-30 points), mild (18-23 points), and severe defect (0-17 points). Patients whose scores were above 24 were included in the study. The instrument had been used by Wu et al., Mazlom et al., Bahrami et al., and Khandare et al. [15, 24, 28, 29].

Based on Brunnstrom stages (the index of motor recovery), patients were divided into six stages according to their legs, arms, and hands function. Stage 1 and stage 6 represent the lowest and highest motor recovery, respectively (Table 1). This tool has already been used by Yavuzer et al., Sütbeyaz et al., and Mazlom et al. [24, 25, 28].

#### Procedure

Initially, according to the inclusion criteria, the Brunnstrom staging tool and MMSE test were completed by the researcher for all patients who were referred to the center. Then, if the score needed to enter the study was obtained, they were introduced to the sampler colleague. Both the intervention groups (MT and the non-reflective surface) and the control group received the usual physiotherapy program, which included neuromuscular electrical stimulation, and then a motor rehabilitation program, which totally lasted for one hour per session. This program was taught to a nurse expert colleague during three 1-hour sessions; thus, at the end of the training, the colleague could correctly explain and perform the exercises in the three study groups. After the usual physiotherapy program for one hour, the intervention groups underwent another rehabilitation program for 30 minutes (15 minutes for the upper limb and 15 minutes for the lower limb) on alternate days in the physiotherapy department during twenty sessions.

The practices used in this study included isotonic active exercise in the range of joints' motion and isometric ones in the front of the mirrors. The colleague exercised the patients during this rehabilitation program. MT was as follows: for the lower limb, the patient was placed in a semi-sitting position on the bed, and the affected leg was placed inside a mirror box measuring 70 by 40 cm so that the patient could not see that leg. The patient was asked to actively move the ankles (dorsiflexion/plantarflexion, eversion/inversion) and knees (flexion/extension) of their unaffected in their range of motion in front of the mirror, while only looking at the image of the intact foot in the mirror and did not receive any verbal feedback during the MT. For the upper limbs, the patient sited on a chair. The affected hand was placed inside a mirror box with dimensions of 35 by 35 cm, in the middle line on the table, and the patient was asked to move his fingers, wrist, and arm in the sagittal plane (flexion/extension), while only paying attention to the movements of the intact hand in the mirror.

The same exercises were performed for the non-reflective surface group, but instead of the mirror, they did these exercises in front of a non-reflective surface, such as a wooden surface, similar in size to the mirror box. The patients looked at the non-reflective surface during the practice (Figure 2). This procedure was repeated in each session. The index of motor recovery of the trained patients

2022, Volume 20, Special Issue on Occupational Therapy



Figure 1. Flow diagram of the study

was re-examined by one of the personnel blinded to the patients who underwent MT before the start of treatment, and at the end of sessions 5, 10, 15, and 20 with the Brunnstrom staging tool [24, 28-30]. Also, the motor recovery of the control group was examined at the mentioned times.

#### Statistical analysis

Types of variables used in this study included continuous numeric (age, duration after stroke, and body mass index) and discrete numeric (muscular strength of the upper limb, muscular strength of the lower limb, minimental state test scores, and motor recovery stages). Categorical-nominal variables included marital status, sex, type of stroke, the side of the body involved, dominant hemisphere, occupation, and the categorical-ordinal variables were education level and monthly income level. Also, in this study, the dependent variable was patients' motor recovery stages, and the independent variable was the type of intervention (MT or non-reflective surface).

The data were analysed by SPSS v. 22. The homogeneity of numeric and categorical variables was confirmed by Analysis of Variance (ANOVA) and Chi-square test, respectively. Also, the patients' MMSE scores homogeneity was evaluated by the Kruskal-Wallis test.

For comparison of Brunnstrom motor recovery stages in each session between different groups, the Kruskal-Wallis test was used. The motor recovery in each group was compared by the Friedman test between different sessions. The distribution of Brunnstrom motor recovery stages before and after the intervention in three study groups was determined and compared at different times by the Kruskal-Wallis test. The groups' pairwise comparison was made by the Mann-Whitney U test.

Moreover, in multiple analyses to determine the effect of MT with controlling the effects of demographic and disease-related variables on motor recovery, GEE (a generalized linear data analysis model) was used and the significance level was considered less than 0.05.

# **3. Results**

All three groups were not significantly different regarding sex, age, body mass index, level of education, monthly income, job, and marital status, and all three groups were ho-



Figure 2. A: Mirror therapy, B: Non-reflective surface

mogeneous. Also, regarding the disease-related variables, including the duration of the stroke, the type of stroke, the affected side, the dominant hemisphere, the muscle strength score of the upper and lower affected limbs, and the MMSE test score, the groups showed no significant difference and all three groups were homogeneous (Table 2).

The trend of change in motor recovery stages in all three study groups was statistically significant, separately (P=0.0001), furthermore, this trend was upward in the non-reflective surface and MT groups, but in the control group based on the mean rank, the trend was downward. Moreover, a comparison of the motor recovery stages in the three study groups separately between different ses-



Iranian Rehabilitation Journal

sions showed no significant difference between the three groups in these sessions. However, it should be noted that in the twentieth session, this difference was nearly significant (P=0.053), and the mean rank of the MT group in terms of motor recovery was higher than the two other groups (Table 3).

There was a significant difference between non-reflective surface and MT groups (P=0.043) in pairwise comparison of the motor recovery stages in the twentieth session, but the difference between the non-reflective surface and control group was not significant. Regarding the results of the twentieth session between the MT and

Stages	Hand	Arm	Leg
1	Flaccidity	Flaccidity	No motor function
2	Gross grasp; very little finger flexion possible	the basic synergies of the limbs mani- fest as weakly associated reactions	A bit voluntary movement; minimal increase in spasticity
3	Gross grasp; grasping objects in hook form without releasing	Maximum spasticity; the voluntary performing of basic limb synergies	Maximum spasticity; the flexion and extension of the hip, knee, and ankle joints' flexion and extension in sitting and standing position
4	The ability of lateral prehension (grasp- ing delicate objects with the thumb and other fingers); a small degree of finger extension and limited thumb move- ments	Decrease in muscle spasticity; the ability to perform combination move- ments deviating from limb synergies	Ninety-degree flexion of the knee in sitting position, while the foot sliding backward on the ground; dorsiflexion of the foot with the heel on the ground & 90-degree flexion of the knee
5	Spherical and cylindrical grasp while releasing possible; palmar prehension	Synergies are not significant anymore; ability to perform more combina- tion movements deviating from limb synergies	Ankle dorsiflexion with the hip and knee extended; knee flexion with the hip extended in the standing position
6	All types of prehension; isolated finger movements in their full range of mo- tion; voluntary flexion of the fingers at all angles	The disappearance of spasticity except while performing rapid movements; isolated joint movement easily pos- sible	Hip abduction in sitting or standing position; bilateral hip internal and exter- nal rotation with ankle inversion and eversion in sitting position

Table 1. Brunnstrom motor recovery stages

Iranian Rehabilitation Journal

Variable		1	Sig		
		Non-Reflective	Mirror Therapy	Control	JIE.
Age (y	ears)	58.8±6.7	57.5±7.5	57±9.1	0.658**
Sev	Male	21(67.7)	20(64.5)	18(58.1)	0 723*
JEA	Female	10(32.3)	11(35.5)	13(41.9)	0.725
Marital status	Single	3(9.7)	3(9.7)	7(22.6)	0 239*
Wanta status	Married	28(90.3)	28(90.3)	24(77.4)	0.235
Body Mass In	dex (kg/m²)	25.53±2.77	25.09±2.38	24.93±3.85	0.725**
	Less than adequate	9(29.0)	9(29.0)	11(35.5)	
Monthly income	Adequate	22(71)	22(71)	20(64.5)	0.818*
	More than adequate	0(0)	0(0)	0(0)	
	Junior High School	14(45.2)	13(41.9)	14(45.2)	
Level of education	High school	15(48.4)	16(51.6)	15(48.4)	0.999*
	Academic	2(6.5)	2(6.5)	2(6.5)	
	Employee	12(38.7)	11(35.5)	10(32.3)	
Job	Self-employment	5(16.1)	10(32.3)	7(22.6)	0.6*
	Housekeeper	14(45.2)	10(32.3)	14(45.2)	
Post-stroke du	ration (years)	5±5	4±5	5±6	0.649**
MMSE score (0-30)		25.87±1.26	25.61±1.65	26±1.24	0.157***
Type of stroke	Ischemic	29(93.5)	27(87.1)	26(83.9)	0.496*
Type of stroke	Hemorrhagic	2(6.5)	4(12.9)	5(16.1)	0.480
Damaged side of the	Right	7(22.6)	27(87.1)	26(83.9)	0.688*
brain	Left	24(77.4)	4(12.9)	5(16.1)	0.088
Dominant homisphore	Right	2(6.5)	1(3.2)	5(16.1)	0.160*
Dominant hemisphere	Left	29(93.5)	30(96.8)	26(83.9)	0.109
	1	0	0	5(16.1)	
Muscular strength score	2	10(32.3)	9(29)	10(32.3)	0.133*
of lower limbs (0-5)	3	18(58.1)	17(54.8)	13(41.9)	
	4	3(9.7)	5(16.1)	3(9.7)	

Table 2. Demographic and disease-related characteristics of the patients (n=93)

\*Chi-square; \*\*ANOVA; \*\*\*Kruskal-Wallis test

Iranian Rehabilitation Journal

Forcione	Stager		Cia *		
565510115	Jiages	Non-reflective	Mirror Therapy	Control	Sig.
	1	4(12.9)	3(9.7)	2(6.5)	
Before the intervention	2	16(51.6)	10(32.3)	14(45.2)	0.237
	3				
Mean Rank		41.24	51.52	48.24	
	1	0(0)	0(0)	1(3.2)	
Socion F	2	5(16.1)	5(16.1)	4(12.9)	0 491
262210112	3	17(54.8)	11(35.5)	13(41.9)	0.461
	4	9(29)	15(48.4)	13(41.9)	
Mean Rank		42.89	50.44	47.68	
	2	1(3.2)	0(0)	1(3.2)	
	3	10(32.3)	8(25.8)	11(35.5)	
Session 10	4	14(45.2)	14(45.2)	10(32.3)	0.524
	5	6(19.4)	8(25.8)	9(29)	
	6	0(0)	1(3.2)	0(0)	
Mean Rank		44.02	51.1	45.89	
	3	3(9.7)	1(3.2)	5(16.1)	
Service 1E	4	13(41.9)	11(35.5)	11(35.5)	0.261
Session 15	5	11(35.5)	12(38.7)	9(29)	0.501
	6	4(12.9)	7(22.6)	6(19.4)	
Mean Rank		44.24	52.35	44.4	
	3	1(3.2)	0(0)	3(9.7)	
Section 20	4	9(29)	6(19.4)	9(29)	0.052
5533011 20	5	14(45.2)	10(32.3)	11(35.5)	0.055
	6	7(22.6)	15(48.4)	8(25.8)	
Mean Rank		43.37	56.02	41.61	
Sig.**		0.001	0.001	0.001	

Table 3. Comparing Brunnstrom stages before and after the intervention between three study groups

\* Kruskal-wallis test; \*\* Friedman test.

control group, there was also a significant statistical difference (P=0.0332) (Table 4).

The motor recovery stages of the twentieth session in the non-reflective and MT groups did not have any significant relationship with any demographic variables, but in the control groups, it showed only a significant difference with the income level (P=0.0029). Furthermore, the motor recovery stages of the twentieth session in the non-reflective surface group did not have any significant relationship with any of the disease-related variables. However, in the MT group, it showed a significant relationship only with the upper limb muscle strength scores (P=0.009), in the control group, the difference was only significant between the lower limb muscle strength scores and motor recovery stages (P=0.04).

Iranian Rehabilitation Journal

GEE model showed that the effect of MT with controlling of the demographic and disease-related variables was not significant on motor recovery. The data showed

	Stages	No.(%)			Sig.			
Session		Non-Reflec- tive	Mirror Therapy	Control	Three Groups <sup>*</sup>	Non- Reflective and MT <sup>**</sup>	Placebo and Control <sup>**</sup>	MT and Control <sup>**</sup>
	3	1(3.2)	0(0)	3(9.7)	3(9.7) 9(29) 0.053 11(35.5)		0.738	0.032
Coordina 20	4	9(29)	6(19.4)	9(29)		0.042		
Session 20	5	14(45.2)	10(32.3)	11(35.5)		0.043		
	6	7(22.6)	15(48.4)	8(25.8)				
Session 20	5 6	14(45.2) 7(22.6)	10(32.3) 15(48.4)	11(35.5) 8(25.8)	0.053	0.043		

Table 4. Pairwise comparison of Brunnstrom stages between study groups

\*Kruskal-Wallis test; \*\*Man-Whitney test

Iranian Rehabilitation Dournal

Table 5. The related factors to motor recovery using the GEE model in the mirror therapy group

		Chandrad France —	95%	<b>C</b> 1	
Groups & Variables	Regression Coefficient	Standard Error	Lower	Upper	Sig.
Non-reflective	-0.112	0.109	-0.326	0.102	0.304
Mirror therapy	0.236	0.1294	-0.018	0.489	0.069
Control	Ref				
Junior high school education	-0.762	0.1761	-1.108	-0.417	0
High school education	-0.689	0.1743	-1.031	-0.347	0
Academic education	Ref				
Inadequate income	-0.295	0.1512	-0.591	0.001	0.051
Adequate income	Ref				
Employee	-0.042	0.0648	-0.169	0.085	0.512
Self-employment	-0.36	0.0762	-0.509	-0.211	0
Housekeeper	Ref				
Age<50 years	-0.348	0.1217	-0.587	-0.11	0.004
50-60 years	0.011	0.0426	-0.073	-0.094	0.801
Age≥60 years	Ref				
Married	-0.39	0.1587	-0.701	-0.079	0.014
Single	Ref				
Right Dominant hemisphere	0.168	0.1773	-0.179	0.516	0.343
Left Dominant hemisphere	Ref				
Body mass index (BMI)	-0.005	0.0013	-0.008	-0.003	0
Upper limb muscle strength	-0.066	0.0197	-0.105	-0.028	0.001
Lower limb muscle strength	0.513	0.1364	0.245	0.78	0

Iranian Rehabilitation Journal

that education level was one of the effective variables on the outcome of MT in motor recovery (P=0.0001). MT was less effective in the group with low income than with high income (P=0.051). There was also a significant relationship between age and the effect of MT on motor recovery (P=0.004) so that it was less effective in patients under 50 than in patients over 60 years old. Also, the effect of the intervention was less in married patients than that in single patients (P=0.014). A significant relationship was found between motor recovery stage and body mass index (P=0.0001), upper limb muscle strength score (P=0.001), and lower limb muscle strength score (P=0.0001). MT was less effective in patients with higher body mass index, and muscular strength of the upper limbs, and this relationship was direct with the muscular strength of the lower limbs (Table 5).

#### 4. Discussion

According to the results of this study, motor recovery stages increased in all three groups of MT, the nonreflective surface, and the control, but the stages of the MT group increased more (P=0.0001). In patients, after stroke, there is some self-improvement in their motor functions, but it is not remarkable, and it will disappear if the patients do not receive physiotherapy programs and adjuvant therapies. By moving the intact limb in the front of the mirror, not only the motor cortex relates the moving limb activates but also the contralateral motor cortex is activated, as a result of the reflection of the movement in the mirror [31].

The functional magnetic resonance imaging proved that MT makes the asymmetry in beta desynchronization (related to the movement) between hemispheres more symmetrical [32]. Although MT has influenced the motor function of the affected organs in all studies, this effect varies in different studies. The results of a study done by Wu et al. showed that MT had significant effects on the motor function of the distal parts of the hand, as the upper limb motor function, measured by the Fogel Meyer tool, increased by 3.7% [15]. In this study, more minor improvement in motor function was reported compared to the present study (3.7% versus 8.7%). Perhaps this difference can be attributed to the selection of patients with chronic stroke, mild spasticity, and mild to moderate motor disability in Wu et al.'s study. However, in the present study, subacute and chronic patients with moderate to severe motor disabilities were examined. Spasticity of the joints was also not investigated.

Yavuzer et al. also reported that the increase in motor function after intervention for the upper limb was 3.8% [25]. Although the duration of MT in our study was half of that in the mentioned study, the effect was almost doubled, perhaps due to differences in how the MT program was done for the upper limb.

In another study by Invernizzi et al., improved motor function after one month of MT in the intervention group showed a significant difference from the control group. The patients' motor function scores, measured by the FIM index, were reported to increase by 31.5% in the MT group; thus, the rate of improvement in motor function was almost 3.5 times higher than our study [33]. This increase of 3.5 times may be due to the longer duration of each session of MT (30 minutes in the first two weeks and 1 hour in the second two weeks) and selecting patients with ischemic stroke in the acute stage (less than four weeks from the onset of stroke).

Thieme et al. also evaluated the effect of MT on motor recovery after stroke in their review article. According to this article, MT had a significant impact on motor function [34]. However, in our study, the motor function improvement in terms of motor recovery index was lower than other studies mentioned by Thieme et al. [34]. This difference may be due to the type of movements performed in MT, the duration of this intervention, the variety of instruments used, and the criteria for selecting the patients, participating in the study.

In another study, Mazlom et al. also illustrated that MT increases patients' motor function after stroke, and 20 sessions of MT caused significant improvement in patients' motor recovery. In the mentioned study and our study, almost equal improvement in motor function was reported (9.7% vs. 8.7%) [24].

Considering the borderline significant relationship of MT with income levels, the results show that in patients with higher income levels, MT was more effective on motor recovery (P=0.051). This effect can be due to the psychological factors caused by higher income and also the greater participation of these patients in the MT program.

Job is another variable that has an effect on motor recovery due to MT (P=0.0001). The effect of MT was higher in patients who were self-employed than housekeepers. Perhaps the reasons are higher concentration and lower anxiety in housekeepers. On the other hand, it may occur due to the practice of MT in homes by housekeepers, which is one of the limitations of our study.

There was also a significant relationship between age and the effect of MT on motor recovery (P=0.004). Age is one of the factors that affect the outcome of this therapy after the stroke. Besides age, the severity of the neurological disorder, cause of stroke, and comorbidities can also affect the prognosis of stroke [35].

In married patients, the effect of the intervention was less than in single patients (P=0.0144). Marriage and social support have been reported to decline depression and increase the quality of life. Married people with chronic illnesses tend to recover more than divorced people. They have better physical and mental conditions, which is due to the social support that married people receive. Also, married people report fewer economic problems [36, 37]. However, in our study, intervention in married patients was less effective, which can be referred to the small number of single patients in our samples or the problems of married patients, especially in our society, like socio-economic problems, which face these problems more than single patients.

# 5. Conclusion

MT is one of the most practical rehabilitation programs for chronic and subacute stroke patients. Our study suggests that MT, besides routine rehabilitation programs, can help the patients recover more quickly after stroke. Also, this study indicated that many other factors can affect the outcome of MT in these patients; thus, further research is needed to determine the effect of each factor. To consider MT as a part of the routine and home-based rehabilitation program, more clinical trials should be done.

#### Limitation and strength

The present study has three limitations. First, the effect of MT was investigated among patients without cognitive impairment; hence, we cannot generalize this rehabilitation program to all stroke patients. Second, in this study, the patients' motor recovery was studied after twenty sessions (each session was thirty minutes); thus, we cannot suggest anything about the effect of more prolonged duration use of MT. At last, the third limitation was that we could not control the exercises patients did at home. These exercises may affect the outcome of MT.

Besides these limitations, controlling demographic and disease-related variables, and also considering three study groups (MT, non-reflective surface, and control) are the advantages of our study. To control the placebo effect of MT, a third group, the non-reflective group, was used in the present study.

# **Ethical Considerations**

#### Compliance with ethical guidelines

The present study was approved by the Ethics Committee of Guilan University of Medical Sciences with the code IR.GUMS.REC.1394.11 and registered at the Iranian Registry of Clinical Trials with the code IRCT201504224787N5.

#### Funding

This work was supported by School of Nursing and Midwifery, Guilan University of Medical Sciences.

#### Authors' contributions

All authors equally contributed to preparing this article.

#### **Conflict of interest**

The authors declared no conflict of interest.

#### Acknowledgments

The authors are grateful to the authorities of the Physiotherapy Center of Disabled and Elderly Hospice and School of Nursing and Midwifery and all the participants in this study.

#### References

- [1] Mozaffarian D, Benjamin EJ, Go AS, Arnett DK, Blaha MJ, Cushman M, et al. Heart disease and stroke statistics-2015 update: A report from the American Heart Association. Circulation. 2015; 131(4):e29-322. [PMID]
- [2] Bennett DA, Krishnamurthi RV, Barker-Collo S, Forouzanfar MH, Naghavi M, Connor M, et al. The global burden of ischemic stroke: Findings of the GBD 2010 study. Global Heart. 2014; 9(1):107-12. [PMID]
- [3] Gandhi DB, Sterba A, Khatter H, Pandian JD. Mirror therapy in stroke rehabilitation: Current perspectives. Therapeutics and Clinical Risk Management. 2020; 16:75-85. [PMID] [PMCID]
- [4] Daneshfard B, Izadi S, Shariat A, Toudaji MA, Beyzavi Z, Niknam L. Epidemiology of stroke in Shiraz, Iran. Iranian Journal of Neurology. 2015; 14(3):158-63. [PMID]
- [5] Truelsen T, Piechowski-Jóźwiak B, Bonita R, Mathers C, Bogousslavsky J, Boysen G. Stroke incidence and prevalence in Europe: A review of available data. European Journal of Neurology. 2006; 13(6):581-98. [PMID]
- [6] Hosseini AA, Sobhani-Rad D, Ghandehari K, Benamer HT. Frequency and clinical patterns of stroke in Iran - Sys-

tematic and critical review. BMC Neurology. 2010; 10:72. [DOI:10.1186/1471-2377-10-72] [PMID] [PMCID]

- [7] Ghandehari K. Epidemiology of stroke in Iran. Galen Medical Journal. 2016; 5(S1):3-9. https://www.gmj.ir/index.php/ gmj/article/view/588
- [8] Arene N, Hidler J. Understanding motor impairment in the paretic lower limb after a stroke: A review of the literature. Topics in Stroke Rehabilitation. 2009; 16(5):346-56. [PMID]
- [9] Costa VdS, Silveira JCCd, Clementino TCA, Borges LRDdM, Melo LPd. Effects of mirror therapy on the motor and functional recovery of post-stroke paretic upper limbs: A systematic review. Fisioterapia e Pesquisa. 2016; 23(4):431-8. [DOI:10.1590/1809-2950/15809523042016.]
- [10] Lim KB, Lee HJ, Yoo J, Yun HJ, Hwang HJ. Efficacy of mirror therapy containing functional tasks in poststroke patients. Annals of Rehabilitation Medicine. 2016; 40(4):629-36. [DOI:10.5535/arm.2016.40.4.629] [PMID] [PMCID]
- [11] Pervane Vural S, Nakipoglu Yuzer GF, Sezgin Ozcan D, Demir Ozbudak S, Ozgirgin N. Effects of mirror therapy in stroke patients with complex regional pain syndrome type 1: A randomized controlled study. Archives of Physical Medicine and Rehabilitation. 2016; 97(4):575-81. [PMID]
- [12] Rajappan R, Abudaheer S, Selvaganapathy K, Gokanadason D. Effect of mirror therapy on hemiparetic upper extremity in subacute stroke patients. International Journal of Physiotherapy. 2015; 2(6):1041-6. [DOI:10.15621/ijphy/2015/ v2i6/80766]
- [13] Hatem SM, Saussez G, Della Faille M, Prist V, Zhang X, Dispa D, et al. Rehabilitation of motor function after stroke: A multiple systematic review focused on techniques to stimulate upper extremity recovery. Frontiers in Human Neuroscience. 2016; 10:442. [PMID] [PMCID]
- [14] Mirela Cristina L, Matei D, Ignat B, Popescu CD. Mirror therapy enhances upper extremity motor recovery in stroke patients. Acta Neurologica Belgica. 2015; 115(4):597-603. [DOI:10.1007/s13760-015-0465-5] [PMID]
- [15] Wu CY, Huang PC, Chen YT, Lin KC, Yang HW. Effects of mirror therapy on motor and sensory recovery in chronic stroke: A randomized controlled trial. Archives of Physical Medicine and Rehabilitation. 2013; 94(6):1023-30. [PMID]
- [16] Ramachandran VS, Rogers-Ramachandran D. Synaesthesia in phantom limbs induced with mirrors. Proceedings. Biological Sciences. 1996; 263(1369):377-86. [PMID]
- [17] Acharya S, Shukla S. Mirror neurons: Enigma of the metaphysical modular brain. Journal of Natural Science, Biology, and Medicine. 2012; 3(2):118-24. [PMID] [PMCID]
- [18] Rassafiani M, Al-Fraij A, Al-Mahri A, Manee F. The effectiveness of mirror therapy on upper limb function in stroke patients: A single case experimental design. Iranian Rehabilitation Journal. 2019; 17(3):285-92. [DOI:10.32598/irj.17.3.285]
- [19] Fuchs E, Flügge G. Adult neuroplasticity: More than 40 years of research. Neural Plasticity. 2014; 2014:541870. [PMID] [PMCID]
- [20] Cacchio A, De Blasis E, De Blasis V, Santilli V, Spacca G. Mirror therapy in complex regional pain syndrome type 1 of the upper limb in stroke patients. Neurorehabilitation and Neural Repair. 2009; 23(8):792-9. [PMID]

- [21] Vladimir Tichelaar YI, Geertzen JH, Keizer D, Paul van Wilgen C. Mirror box therapy added to cognitive behavioural therapy in three chronic complex regional pain syndrome type I patients: A pilot study. International Journal of Rehabilitation Research. 2007; 30(2):181-8. [PMID]
- [22] Foell J, Bekrater-Bodmann R, Diers M, Flor H. Mirror therapy for phantom limb pain: Brain changes and the role of body representation. European Journal of Pain. 2014; 18(5):729-39. [PMID]
- [23] Kim SY, Kim YY. Mirror therapy for phantom limb pain. The Korean Journal of Pain. 2012; 25(4):272-4. [DOI:10.3344/ kjp.2012.25.4.272] [PMID] [PMCID]
- [24] Mazlom S, bahrami M, Hasanzadeh F, Ghandehari K. [The effect of mirror therapy on motor abilities of patients with stroke (Persian)]. Evidence Based Care Journal. 2013; 3(3):7-18. [DOI:10.22038/EBCJ.2013.1566]
- [25] Yavuzer G, Selles R, Sezer N, Sütbeyaz S, Bussmann JB, Köseoğlu F, et al. Mirror therapy improves hand function in subacute stroke: A randomized controlled trial. Archives of Physical Medicine and Rehabilitation. 2008; 89(3):393-8. [PMID]
- [26] Sütbeyaz S, Yavuzer G, Sezer N, Koseoglu BF. Mirror therapy enhances lower-extremity motor recovery and motor functioning after stroke: A randomized controlled trial. Archives of Physical Medicine and Rehabilitation. 2007; 88(5):555-9. [PMID]
- [27] Dohle C, Püllen J, Nakaten A, Küst J, Rietz C, Karbe H. Mirror therapy promotes recovery from severe hemiparesis: A randomized controlled trial. Neurorehabilitation and Neural Repair. 2008; 23(3):209-17. [DOI:10.1177/1545968308324786] [PMID]
- [28] Bahrami M, Mazloom SR, Hasanzadeh F, Ghandehari K. [The effect of mirror therapy on self care of patients with stroke (Persian)]. Journal of Mazandaran University of Medical Sciences. 2013; 23(107):78-88. http://jmums.mazums. ac.ir/article-1-2979-en.html
- [29] Khandare SS, Singaravelan R, Khatri SM. Comparison of task specific exercises and mirror therapy to improve upper limb function in subacute stroke patients. IOSR Journal of Dental and Medical Sciences. 2013; 7(1):5-14. [DOI:10.9790/0853-0710514.]
- [30] Houglum PA, Bertoti DB. Biographical Sketch of Signe Brunnstrom, 1898-1988. In: Houglum PA, Bertoti DB, editors. Brunnstrom's clinical kinesiology. Pennsylvania: Davis Company; 2012. https://www.google.com/books/edition/ Brunnstrom\_s\_Clinical\_Kinesiology/rX1EYgEACAAJ?hl=en
- [31] Arya KN. Underlying neural mechanisms of mirror therapy: Implications for motor rehabilitation in stroke. Neurology India. 2016; 64(1):38-44. [PMID]
- [32] Rossiter HE, Borrelli MR, Borchert RJ, Bradbury D, Ward NS. Cortical mechanisms of mirror therapy after stroke. Neurorehabilitation and Neural Repair. 2015; 29(5):444-52. [DOI:10.1177/1545968314554622] [PMID]
- [33] Invernizzi M, Negrini S, Carda S, Lanzotti L, Cisari C, Baricich A. The value of adding mirror therapy for upper limb motor recovery of subacute stroke patients: A randomized controlled trial. European Journal of Physical and Rehabilitation Medicine. 2013; 49(3):311-7. [PMID]
- [34] Thieme H, Morkisch N, Mehrholz J, Pohl M, Behrens J, Borgetto B, et al. Mirror therapy for improving motor function

after stroke. The Cochrane Database of Systematic Reviews. 2018; 7(7):CD008449. [PMID] [PMCID]

- [35] Alawieh A, Zhao J, Feng W. Factors affecting post-stroke motor recovery: Implications on neurotherapy after brain injury. Behavioural Brain Research. 2018; 340:94-101. [PMID] [PM-CID]
- [36] Jafari M, Makarem A, Dalvandi A, Azimian M, Hosseini MA. [Determination of facilitators and barriers in post stroke life, in Kerman city (Persian)]. Journal of Modern Rehabilitation. 2011; 5(2):54-62. http://mrj.tums.ac.ir/article-1-77-en.html
- [37] Beckett J, Barley J, Ellis C. Patient perspectives of barriers and facilitators of treatment-seeking behaviors for stroke care. Journal of Neuroscience Nursing. 2015; 47(3):154-9. [DOI:10.1097/JNN.00000000000134] [PMID]

This Page Intentionally Left Blank